Author’s response to reviews

Title: Physical growth: is it a good indicator of development in early childhood in low- and middle-income countries?

Authors:
Thach Tran (thach.tran@monash.edu)
Sara Holton (sara.holton@monash.edu)
Hau Nguyen (hau.nguyen@monash.edu)
Jane Fisher (jane.fisher@monash.edu)

Version: 2 Date: 19 Feb 2019

Author’s response to reviews:

Dear Editorial Office,

Re: Manuscript entitled “Physical growth: is it a good indicator of development in early childhood in low- and middle-income countries?”

We are really grateful for your appraisal and the reviewers’ thoughtful comments. We have addressed all comments. We hope our manuscript can now be considered for publication.

Yours sincerely,

On behalf of the authors

Thach Tran